 Vc/22-Nov-04
Week48 starting 22 Nov 04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 22 Nov 04  ~11/22/2004 12:38:45 PM 
	LHC SM18 Magnet Test Facility 


	5
	A1
	MBBL3219
	Prep 5 Pumping + Cool Down
	90.2

	9
	A2
	MBAR3204
	ICS 6 Disconnect Magnet WP07
	299.1

	12
	B1
	SSS614
	ICS 5 Traveller Last
	295.6

	1
	B2
	MBBL2043
	PTE 1 Minimum NRJ + RRR
	9.8

	7
	C1
	MBAR3257
	PT 12 Warm Up
	79.5

	4
	C2
	MBBL1168
	Prep 5 Pumping + Cool Down
	294.6

	11
	D1
	SSS039
	CL-1 Call Transport
	295.1

	2
	D2
	SSS101
	PT 6.2 Training 2
	1.9

	6
	E1
	MBBR3259
	Prep 5 Pumping + Cool Down
	148.4

	10
	E2
	MBBL3237
	Finished Ready to go Final scan
	295.4

	8
	F1
	MBBL3216
	Prep 5 Pumping + Cool Down
	294.2

	3
	F2
	MBAL3225
	PT 6.6 Training 6
	2.5 HV


   Temperature data are taken from TBSM18 site
Coordinator this week: Vittorio; MTM support on call :   16-3602 .

15 named  Magnets Tested or 16 Magnet  tests  cycles:

 13 dipoles( 2 with MM)+ 2 SSS( 0 with MM ) +  0  SSSS     Disconnected after tests last week ( Sat Midnight to Sat midnight) 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
AOB

Tests 

	Magnets under Test from 14-11-2004 to 20-11-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAR1134 
	[image: image1.png]



	10-11-2004 
	23 h.
	30 h.
	25 h.
	14 h.
	8 h.
	14-11-2004 
	96 hours
	
	

	2
	TBA1
	MBBL3220 
	[image: image2.png]



	14-11-2004 
	28 h.
	43 h.
	31 h.
	15 h.
	13 h.
	20-11-2004 
	144 hours
	
	

	3
	TBA2
	MBBL2112 
	[image: image3.png]



	11-11-2004 
	30 h.
	55 h.
	27 h.
	15 h.
	6 h.
	17-11-2004 
	144 hours
	
	

	4
	TBB1
	SSS042 
	[image: image4.png]



	06-11-2004 
	38 h.
	62 h.
	111 h.
	-
	-
	15-11-2004 
	216 hours
	
	

	5
	TBB1
	SSS042 
	[image: image5.png]



	15-11-2004 
	0 h.
	15 h.
	17 h.
	10 h.
	9 h.
	18-11-2004 
	72 hours
	
	

	6
	TBB2
	MBBR3241 
	[image: image6.png]



	09-11-2004 
	44 h.
	60 h.
	20 h.
	22 h.
	7 h.
	16-11-2004 
	168 hours
	
	

	7
	TBC1
	MBAR1138 
	[image: image7.png]



	11-11-2004 
	38 h.
	53 h.
	55 h.
	22 h.
	7 h.
	18-11-2004 
	168 hours
	
	

	8
	TBC2
	MBAR3200 
	[image: image8.png]



	08-11-2004 
	21 h.
	46 h.
	46 h.
	14 h.
	4 h.
	14-11-2004 
	144 hours
	
	

	9
	TBC2
	MBBL2119 
	[image: image9.png]



	14-11-2004 
	24 h.
	51 h.
	19 h.
	25 h.
	10 h.
	19-11-2004 
	120 hours
	
	

	10
	TBD2
	SSS071 
	[image: image10.png]



	04-11-2004 
	149 h.
	35 h.
	83 h.
	8 h.
	10 h.
	17-11-2004 
	312 hours
	
	

	11
	TBE1
	MBAL2117 
	[image: image11.png]



	15-11-2004 
	25 h.
	44 h.
	21 h.
	15 h.
	12 h.
	20-11-2004 
	120 hours
	
	

	12
	TBE1
	MBBL1164 
	[image: image12.png]



	10-11-2004 
	28 h.
	65 h.
	15 h.
	18 h.
	5 h.
	15-11-2004 
	120 hours
	
	

	13
	TBE2
	MBAL2097 
	[image: image13.png]



	12-11-2004 
	0 h.
	37 h.
	28 h.
	23 h.
	4 h.
	16-11-2004 
	96 hours
	
	

	14
	TBF1
	MBBL1147 
	[image: image14.png]



	11-11-2004 
	53 h.
	43 h.
	22 h.
	31 h.
	13 h.
	18-11-2004 
	168 hours
	
	

	15
	TBF1
	MBBL1167 
	[image: image15.png]



	18-11-2004 
	-
	-
	-
	-
	-
	20-11-2004 
	48 hours
	
	

	16
	TBF2
	MBAL1159 
	[image: image16.png]



	08-11-2004 
	29 h.
	48 h.
	47 h.
	15 h.
	7 h.
	14-11-2004 
	144 hours
	
	

	17
	TBF2
	MBBR3244 
	[image: image17.png]



	14-11-2004 
	25 h.
	32 h.
	21 h.
	13 h.
	14 h.
	19-11-2004 
	120 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR1134
	12046 A.
(D2-L-CFB)
	12362 A.
(D1-U-MRB)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3220
	11474 A.
(D2_U-MRB)
	12024 A.
(D2_U_CFB)
	12284 A.
(D2_U)
	12406 A.
(D1-U-CF4)
	12576 A.
(D2_L_CF_)
	[image: image19.png]



	-
	-
	-
	-
	-
	-
	-

	3
	MBBL2112
	10970 A.
(D2-U)
	12033 A.
(D2-L)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS042
	10855 A.
(QE-12)
	11105 A.
(QI-34)
	11856 A.
(QE-34)
	11998 A.
(QE-12)
	11987 A.
(QI_34)
	12291 A.
(QI_34)
	12452 A.
(QI-34)
	12674 A.
(QI-34)
	12325 A.
(QI_12)
	[image: image21.png]



	-
	-
	-

	5
	SSS042
	12199 A.
(QE-12)
	12507 A.
(QE-34)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBBR3241
	11859 A.
(D2_U)
	11387 A.
(D2-U)
	12850 A.
[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBAR1138
	11500 A.
(D1-L-S01)
	12025 A.
(D2-L-S13)
	12172 A.
(D1_U_S13)
	[image: image24.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBAR3200
	9429 A.
(D2_L)
	10992 A.
(D2_L)
	11368 A.
(D2_L)
	11874 A.
(D1-U)
	12246 A.
(D1 L)
	12340 A.
(D2-L)
	12630 A.
(D1_L)
	12850 A.
[image: image25.jpg]



	-
	-
	-
	-
	-

	9
	MBBL2119
	11429 A.
(D1-U)
	12503 A.
(D1-U)
	[image: image26.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS071
	12480 A.
(QE_12)
	12632 A.
(QI_34)
	12850 A.
[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBAL2117
	11851 A.
(D2_U)
	12489 A.
(D2-U)
	[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBL1164
	11885 A.
(D1_L)
	12295 A.
(D1-L)
	[image: image29.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL2097
	12700 A.
(D2-L)
	12850 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBL1147
	12060 A.
(D2_U)
	12284 A.
(D2_U)
	[image: image31.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	15
	MBBL1167
	[image: image32.jpg]



	-REMOVED: CFB LEAK
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	16
	MBAL1159
	11103 A.
(D2 L)
	11815 A.
(D1-L)
	12017 A.
(D1-L)
	12365 A.
(D2-L)
	12425 A.
(D1-L)
	12414 A.
(D1_L)
	12567 A.
(D1_U)
	12730 A.
(D1-L)
	12846 A.
(D2_U)
	[image: image33.png]



	-
	-
	-

	17
	MBBR3244
	12808 A.
(D2-L)
	12648 A.
(D2-L)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	
	
	
	
	
	
	
	
	
	
	
	
	
	
	

	From 14-11-2004 to 20-11-2004 

	-
	Bench
	Magnet Name
	shafts
	Start Date .
	Connect
Time
	Wait
+Cooling
	Cold
Tests
	Wait 
+Warming
	Disconnect
Time
	End Date .
	Total
	Q

	1
	TBF2
	MBAL1159 
	False 
	08-11-2004 
	29 h.
	48 h.
	47 h.
	15 h.
	7 h.
	14-11-2004 
	6 days
	9 

	2
	TBF2
	MBBR3244 
	False 
	14-11-2004 
	25 h.
	32 h.
	21 h.
	13 h.
	14 h.
	19-11-2004 
	5 days
	2 

	3
	TBF1
	MBBL1147 
	False 
	11-11-2004 
	53 h.
	43 h.
	22 h.
	31 h.
	13 h.
	18-11-2004 
	7 days
	2 

	4
	TBF1
	MBBL1167 
	False 
	18-11-2004 
	98 h.
	0 h.
	0 h.
	0 h.
	-57 h.
	20-11-2004 
	2 days
	0 

	5
	TBE2
	MBAL2097 
	False 
	12-11-2004 
	0 h.
	37 h.
	28 h.
	23 h.
	4 h.
	16-11-2004 
	4 days
	1 

	6
	TBE1
	MBAL2117 
	False 
	15-11-2004 
	25 h.
	44 h.
	21 h.
	15 h.
	12 h.
	20-11-2004 
	5 days
	2 

	7
	TBE1
	MBBL1164 
	False 
	10-11-2004 
	28 h.
	65 h.
	15 h.
	18 h.
	5 h.
	15-11-2004 
	5 days
	2 

	8
	TBD2
	SSS071 
	False 
	04-11-2004 
	149 h.
	35 h.
	83 h.
	8 h.
	10 h.
	17-11-2004 
	13 days
	2 

	9
	TBC2
	MBAR3200 
	False 
	08-11-2004 
	21 h.
	46 h.
	46 h.
	14 h.
	4 h.
	14-11-2004 
	6 days
	7 

	10
	TBC2
	MBBL2119 
	False 
	14-11-2004 
	24 h.
	51 h.
	19 h.
	25 h.
	10 h.
	19-11-2004 
	5 days
	2 

	11
	TBC1
	MBAR1138 
	True 
	11-11-2004 
	38 h.
	53 h.
	55 h.
	22 h.
	7 h.
	18-11-2004 
	7 days
	3 

	12
	TBB2
	MBBR3241 
	False 
	09-11-2004 
	44 h.
	60 h.
	20 h.
	22 h.
	7 h.
	16-11-2004 
	7 days
	2 

	13
	TBB1
	SSS042 
	False 
	06-11-2004 
	38 h.
	62 h.
	111 h.
	169 h.
	-169 h.
	15-11-2004 
	9 days
	9 

	14
	TBB1
	SSS042 
	False 
	15-11-2004 
	0 h.
	15 h.
	17 h.
	10 h.
	9 h.
	18-11-2004 
	3 days
	2 

	15
	TBA2
	MBAR3204 
	False 
	17-11-2004 
	46 h.
	26 h.
	23 h.
	14 h.
	1 h.
	18-11-2004 
	1 days
	2 

	16
	TBA2
	MBBL2112 
	False 
	11-11-2004 
	30 h.
	55 h.
	27 h.
	15 h.
	6 h.
	17-11-2004 
	6 days
	2 

	17
	TBA1
	MBAR1134 
	False 
	10-11-2004 
	23 h.
	30 h.
	25 h.
	14 h.
	8 h.
	14-11-2004 
	4 days
	2 

	18
	TBA1
	MBBL3220 
	True 
	14-11-2004 
	28 h.
	43 h.
	31 h.
	15 h.
	13 h.
	20-11-2004 
	6 days
	5 


 

 

from 13-11-2004 to 20-11-2004
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	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _16_A


	All benches
	 Temperature is increasing.


	ACR
	
	Interlock problem. Pumping stopped.

	Tuesday _16_N


	TBE2
	MTF not accessible with the mobic


	TS
	MTF
	OK ½ hour after.

	Tuesday _16_N


	TBD1
	Qloc error during IAP


	MTM
	IAP
	

	Tuesday _16_N


	TBD1
	Provoked quench HF at 1,5 kA performed 2 times but no discharge of QH 3 and 4.


	MTM
	
	TC contacted. He decided to increase the quench current ( 2k A, 3 kA)



	Wednesday _17_N


	TBF2
	Detraining occurred in 3244 (12808 A then 12648)
	MTM
	
	TC decided to perform a third one.

	Wednesday _17_N


	TBD1
	MQ checks for MM with shafts. End field polarity errors
	MTM
	Display
	Guy contacted. He advised to proceed. Color code adapted to the dipole. Changes will be done during the shut down.

	Thursday _18_M
	TBA2
	Quench localisation with LQA not coherent for the apertures with VTaps
	MTM
	
	Guy and Maryline checked the connection. Discussion with M.Calvi. A different logic in the connection was applied. So the connectors ap1-ap2 were inverted in the CFB side. Corrected. Now cables are labelled.

	Thursday _18_N
	TBF1
	HV test could not be performed: “megger could not be put in remote mode”.
	MTM
	
	Repeated 3 times then rack changed.

	Friday_19_A
	TBF2
	Cryo leak detected in the 3225. (leak in one foot of the magnet, joint not good.)
	ICS?
	
	NC opened by L.Herblin. ACR decided to still go on  with the tests. 

	Friday_19_N
	TBE2
	C11 connector damaged
	
	
	Proceed with the tests but with special care in the plug in and plug out.

	Saturday_20_M
	TBE2
	SPA occurred during the ramp to 12000 A for fast de-excitation test.
	?
	
	Origin not clear?

	Saturday_20_M
	TBD1
	Wire broken during roll angle measurements
	MTM
	
	

	Saturday_20_A
	TBA2
	Water leak in the power cable.
	MTM
	
	Guy came: he checked and advised to proceed because it was not critical but asked to stop the water flow after the power test. Repaired  by Giorgio Monday.

	Saturday_20_A
	TBD1
	HV test after power test. Orbit dipole vs ground failed.
	MTM
	HV
	OK when the step was repeated but failed in the third attempt. Repeated a fourth time then OK.

	Saturday_20_N
	TBB2
	When the shaft 22 was removed, one sector joint was found to be unstuck.
	MTM
	
	Default not detected during measurements, connector and cables not damaged. Shaft shocked during removal?

 Stephane ask to put the shaft out of the circuit. The sector will be electrically checked and the sector repaired (Action Olaf and his team).

	Sunday_21_A
	TBC1
	Incorrect display of the T821 value : 80 K and the magnet is at 1.9 K
	MTM
	
	

	Sunday_21_N
	TBB1
	Leak found on the SSS 614 (localised in a tube  inside the magnet on MRB side)
	
	
	NC will be opened. SSSS sent back?

	Sunday_21_N
	TBF1
	Leak found on the TBF1 test bench (3216 MB) coming from one “flexible” inside the CFB.
	ACR
	
	L.Herblin aware. ACR team changed the flexible Monday.

	Sunday_21_N
	TBA2
	Current lead B overheated. Banana plug damaged.
	
	
	Stéphane informed Guy Monday morning.

	Monday_22_M
	TBB1
	Connector of the cable 16 in the XMT.95 HV rack (N 4) is broken.
	MTM
	
	Guy informed. This rack is out of use for the moment. He will inform Richard for the repair Wednesday.


 









